Author’s response to reviews

Title: "We are the ones who should make the decision" - Knowledge and understanding of the rights-based approach to maternity care

Authors:
Yasmin Jolly (ybjolly@doctors.org.uk)
Mamuda Aminu (Mamuda.Aminu@lstmed.ac.uk)
Florence Mgawadere (Florence.Mgawadere@lstmed.ac.uk)
Nynke van den Broek (Nynke.vandenBroek@lstmed.ac.uk)

Version: 4 Date: 03 Jan 2019

Author’s response to reviews:

1. Please ensure that ALL authors email addresses are included on the title page of your manuscript.

Author emails have been added to the title page as requested.

2. Please provide a Keywords section after your Abstract that includes three to ten keywords representing the main content of the article.

A key words section has been added.

3. Please format your manuscript to include the following subheadings: Background, Methods, Results, Conclusions.

The subheadings have been included as requested.
4. Thank you for providing a consent statement in the Methods section. Please be sure to include a consent statement in the Ethical Approval and consent to Participate section. Please also ensure to clarify the form of consent obtained, as we note that verbal consent was indicated as the form of consent in the Methods, and written consent was noted in the Consent to publish section.

Written consent was obtained and this has been clarified in both sections.

- Please note that the Consent for publication statement refers to consent obtained to include any individual person’s data in any form (including individual details, images or videos). Therefore, please clarify whether consent for publication was obtained. If not, you may state "Not applicable" in this section.

We have revised this to Not applicable as the MS does not contain personal data or images. All included data has been anonymized.

5. In the funding section, please also state the role of the funding body in the design of the study, collection, analysis, interpretation of data, and in writing the manuscript.

We have clarified the role of the funding body.

6. Please upload English language versions of all interview/ focus group guides developed specifically for use in this study as supplementary files. If the interview/ focus group guides are not yours or are already published by you, please supply references and/or links to them.

English versions of the topic guides have been uploaded as supplementary files.

7. Thank you for including your response to reviewers and the tracked changes version of your manuscript as supplementary files, however these files are no longer needed and should be removed from the file inventory.

These files have been removed from the submission.
8. Please ensure that when you upload your revised submission that it is in the final form for publication. Please remove any tracked changes, colored text, or highlighting and include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only.

We have uploaded a single clean copy of the MS.